

## Permits

---

**From:** ORS Administrator <lwbors@yk.com>  
**Sent:** Thursday, June 30, 2016 9:50 AM  
**To:** screeningofficer@eastarm.com; boydw@bathurstinlet.com; barryt@arcticsafaris.ca; dave.white@aurorageosciences.com; christian.bertelsen@cannor.gc.ca; Matthew.Spence@cannor.gc.ca; mauge@yellowknife.ca; rmakohoniuk@yellowknife.ca; walexander@yellowknife.ca; lands@denenation.com; admin\_dkfn@northwestel.net; ima\_dkfn@northwestel.net; craigecologynorth@gmail.com; ec.ea.nwt.ec@canada.ca; fisheriesprotection@dfo-mpo.gc.ca; fieldworker.frmc53@northwestel.net; frmc@northwestel.net; forsmithmetisCouncil@northwestel.net; Alexis\_Campbell@gov.nt.ca; Jon\_Posynick@gov.nt.ca; rhonda\_batchelor@gov.nt.ca; stu\_niven@gov.nt.ca; bill\_pain@gov.nt.ca; Diep\_Duong@gov.nt.ca; gnwt\_ea@gov.nt.ca; Heather\_Beck@gov.nt.ca; kate\_witherly@gov.nt.ca; laurie\_mcgregor@gov.nt.ca; monica\_wendt@gov.nt.ca; Nathen\_Richea@gov.nt.ca; patrick\_clancy@gov.nt.ca; Paul\_Green@gov.nt.ca; Robert\_Jenkins@gov.nt.ca; tyanna\_steinwand@gov.nt.ca; Wendy\_Bidwell@gov.nt.ca; Colin\_merz@gov.nt.ca; doug\_carr@gov.nt.ca; Jeremy\_Roberts@gov.nt.ca; Steven\_Shen@gov.nt.ca; Catherine\_BraunRodriguez@gov.nt.ca; johnny\_lennie@gov.nt.ca; Russell\_Teed@gov.nt.ca; Brandon\_Bradbury@gov.nt.ca; clint\_ambrose@gov.nt.ca; Erin\_Anderson@gov.nt.ca; Jayda\_Robillard@gov.nt.ca; joe\_heron@gov.nt.ca; Lindsay\_Armer@gov.nt.ca; Mike\_Vassal@gov.nt.ca; Norman\_McCowan@gov.nt.ca; pat\_knutson@gov.nt.ca; Scott\_Stewart@gov.nt.ca; Iqbal\_Arshad@gov.nt.ca; jean\_soucy@gov.nt.ca; mark\_davy@gov.nt.ca; Olivia\_Lee@gov.nt.ca; Glen\_Mackay@gov.nt.ca; Pauline\_DeJong@gov.nt.ca; Hilary\_machtans@golder.com; Tasha\_Hall@golder.com; david.alexander@cannor.gc.ca; Carolc.lands@gmail.com; hrmc@northwestel.net; CAU-UCA@aandc.gc.ca; NTCARD@aandc.gc.ca; Sam.Kennedy@aandc.gc.ca; tim.morton@aandc.gc.ca; landsnresources@katlodeeche.com; chief.lkdfn@gmail.com; lkdfnlands@gmail.com; preliminaryscreening@reviewboard.ca; ww.symbion@shawbiz.ca; Heather Scott; Jen Potten; Permits; Gary.Woo@neb-one.gc.ca; shin.shiga@nsma.net; president.nwtmn@northwestel.net; rcc.nwtmn@northwestel.net; Ceo.srfn@northwestel.net; chief.srfn@northwestel.net; chief@slfn196.com; lands@slfn196.com; exec@slema.ca; zabeynevitt@tlichoc.com; jhood@fortsmith.ca; kmorrison@fortsmith.ca; sao@hayriver.com; land@wpfn.ca; Susan.Abernethy@wscc.nt.ca; apower@ykdene.com; esangris@ykdene.com; jblack@ykdene.com  
**Cc:** Erica Janes  
**Subject:** MV2011L3-0001 - Town of Fort Smith - Extension for response to reviewer comments

Good morning, The Town of Fort Smith has requested an extension, which has been granted, to respond to reviewer comments on the Landfill Groundwater Monitoring and Protection Program and the Landfill Interim Closure and Reclamation Plan. The new deadline is July 29, 2016.